Author's response to reviews

Title: Combined mutations of ASXL1, CBL, FLT3, IDH1, IDH2, JAK2, KRAS, NPM1, NRAS, RUNX1, TET2 and WT1 genes in myelodysplastic syndromes and acute myeloid leukemias

Authors:

Julien Rocquain (rocquainju@marseille.fnclcc.fr)
Nadine Carbuccia (carbuccian@marseille.fnclcc.fr)
Virginie Trouplin (trouplinv@marseille.fnclcc.fr)
Stéphane Raynaud (raunauds@marseille.fnclcc.fr)
Anne Murati (muratia@marseille.fnclcc.fr)
Meyer Nezri (Nezri.mrdint@ch-martigues.fr)
Zoulika Tadrist (zoulika.tadrist@ch-salon.fr)
Sylviane Olschwang (olschwangs@marseille.fnclcc.fr)
Norbert Vey (vey@marseille.fnclcc.fr)
Daniel Birbaum (daniel.birbaum@inserm.fr)
Veronique Gelsi-Boyer (gelsiv@marseille.fnclcc.fr)
Marie-Joelle Mozziconacci (mozziconaccimj@marseille.fnclcc.fr)

Version: 6 Date: 27 July 2010

Author's response to reviews:

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.

In the submission system, it is impossible to have the names of the genes in italics.

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.

Corrected.

Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system. Currently they do not match.

Corrected.

Tables: table 1 has been divided into parts. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

Tables: table 2 has been divided into parts. It is important that tables are
numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

We don't know what to do. We HAD the tables in one single part BUT in landscape format. You ask for portrait format, the consequence of which is that the tables are cut into several parts. We don't know how to number that. We'd prefer one single part.

Please help.